Author’s response to reviews

Title: Treatment outcomes and risk factors of extra-pulmonary tuberculosis in patients with comorbidities

Authors:

Amer Hayat Khan (dramer2006@gmail.com)
Syed Azhar Syed Sulaiman (sazhar@usm.my)
Madeeha Laghari (madeehalaghari@gmail.com)
Mohamed Azmi Ahmad Hassali (azmihassali@usm.my)
Abdul Razak Muttalif (drarm28@yahoo.com)
Zohra Bhatti (zohrabhatti@gmail.com)
Long Chiau Ming (long@puncakalam.uitm.edu.my)

Version: 1 Date: 13 Aug 2018

Author’s response to reviews:

1. Please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. If verbal, please state the reason and whether the ethics committee approved this procedure. If the data was de-identified and hence consent was waived by an IRB or if consent is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.
Reply: consent was waived due to retrospective nature of study.

2. If the dataset is private, please provide clarification on whether any (administrative) permission was required, and obtained, in order to review patient records and use the data mentioned in the Methods. Please also state whether the data analysed had been de-identified. Please include this information in the “Ethics approval and consent to participate” section of the Declarations.
Reply: Data was de-identified and mentioned in the manuscript.